Author's response to reviews

Title: Impact of adjuvant chemotherapy for gliomatosis cerebri

Authors:

Doo-Sik Kong (kds026@skku.edu)
Sung Tae Kim (stkim@skku.edu)
Jung-II Lee (jilee@skku.edu)
Yeon-Lim Suh (sis@skku.edu)
Do Hoon Lim (dhlim@skku.edu)
Won Seog Kim (wskim@skku.edu)
Ki-Hoon Kwon (kkw@skku.edu)
Kwan Park (kwanpark@skku.edu)
Jong Hyun Kim (jonghkim@skku.edu)
Do-Hyun Nam (nsnam@skku.edu)

Version: 2 Date: 5 March 2010

Author's response to reviews:

Impact of adjuvant chemotherapy for gliomatosis cerebri

Dear Editor,

We were pleased to hear that BMC Cancer is interested in evaluating our manuscript. We are thankful for the very constructive comments that we received on the previous submission. On behalf of the co-authors of the submitted manuscript Version 1 ["Impact of adjuvant chemotherapy for gliomatosis cerebri"], I have the pleasure of submitting a revised form of this work for consideration.

Based on the reviewer’s comments and suggestions, the manuscript underwent revision and several points in the text have been rewritten, amended, or further analyzed for reasons of precision and clarity. Together with the revised version of the manuscript we are submitting our point-by-point response to the reviewer’s comments.

Author Comments

In the next paragraphs you will find the reviewer’s comments upon the originally submitted version of the manuscript. After each one of these notes you may find our response.

We would like to thank the reviewers for their helpful decisive comments and suggestions that gave us the opportunity of critically revising our work. We hope this revised version of the manuscript has substantially improved to meet the journal’s standards for publication.

Sincerely,
Do-Hyun Nam, M.D., Ph.D.
Corresponding Author